Author’s response to reviews

Title: Regression of solid breast tumours in mice by Newcastle disease virus is associated with production of apoptosis related-cytokines

Authors:
Raihan Juraimi (juraimi@moh.gov.my)
Umar Ahmad (babasarak@yahoo.co.uk)
Yong Keong (yongyoke@upm.edu.my)
Zolkapli Eshak (zolkapli@utm.edu.my)
Fauziah Othman (fauziah@upm.edu.my)
Aini Ideris (aiini@upm.edu.my)

Version: 3 Date: 08 Mar 2019

Author’s response to reviews:
Response to Reviewers’/Editor’s Comments

BCAN-D-17-02230R1
Regression of solid breast tumours in mice by Newcastle disease virus is associated with production of apoptosis related-cytokines

Raihan Juraimi, BSc, MSc; Umar Ahmad, BSc, MS; Yong Yoke Keong, BSc, PhD; Zolkapli Eshak, BSc, MSc, PhD; Fauziah Othman, DVM, MSc, PhD; Aini Ideris, DVM, MSc, PhD

BMC Cancer

Editor’s Comments:

Comment 1: To clarify the points made previously by reviewer 1 regarding your Figures, please add p-value bars and concentration on the y-axis.
Response: Thank you for the comment. We add the p-value bars and the concentration on the y-axis as suggested by reviewer 1 and updated in the new version of the manuscript. This has significantly improved our manuscript.